Author's response to reviews

Title: Measles mimicking HIV seroconversion syndrome: a case report

Authors:

Mahua T Chatterjee (mahua142003@yahoo.co.uk)
Margaret Coleman (megcoleman@doctors.org.uk)
Gary Brook (gary.brook@nwlh.nhs.uk)
Daniel McCrea (daniel.mccrea@nwlh.nhs.uk)

Version: 4 Date: 21 January 2010

Author's response to reviews: see over
Dear Editor,

A revised submission of the medical case report ‘Measles mimicking HIV seroconversion syndrome: a case report’ is attached. MS ID: 9290970952290951

The changes requested have been made as follows:

Major revisions
- Authors list: listed in the preferred style for the journal (first name, Initial, last name)
- Abstract has been structured in the following way: Introduction, Case presentation, Conclusion
- Authors’ contributions have been included
- References are cited at least once through the text. Reference 7 has been removed (this was included originally in error)

Minor revisions
- The following text has been deleted from the title page: ‘Title page, names of authors, Postal address of all affiliations (1,2,3,4), E-mail addresses of all authors’
- The country has been included in the affiliation details
- Figures: The title has been removed from the image file
- The term et al has been removed from the reference list and the list now contains all authors (up to 30)
- The main text is structured: Introduction, Case presentation, Discussion, Conclusion, Consent, Competing interests, Authors’ contributions, References, Figure legends
- Typographical errors have been checked for

Yours faithfully,

Dr Mahua Chatterjee
BSc, PhD, MRCP (UK)

Mahua142003@yahoo.co.uk